

Dear Editor,

Please find the revised manuscript. All the issues from editor are addressed and highlighted in the manuscript. We have a native English speaker to check the English grammar and Track Changes are made for correction. All the issues of review comments have been addressed. Below is the review comments and our responses, detailed on an item-by-item basis. Now, the manuscript shows a lot of Track Changes and highlights, please let me know if a clean version needed and I can convert to a clean version.

Review 1:

This paper reported "Follicular dendritic cell sarcoma detected in hepatogastric ligament". The manuscript is well presented and of interest and can contribute to knowledge of this topic. But minor issues should be corrected.

1. Some grammatical errors are present which should be corrected as well as some medical terminologies.

Thanks! A native English speaker has checked the English grammar throughout the manuscript and made corrections accordingly.

2. Introduction is too short. The statistics about non popularity of the case and the necessity of the case presentation as case report should be noted in the Introduction.

Thanks! The information in term of popularity of the case and the necessity of the case presentation is added.

Review 2:

This is an interesting case report. It can be accepted.

Thanks!

Review 3:

This manuscript is a case report about a very rare localization of Follicular Dendritic Cell Sarcoma (FDSC). I recommend some corrections;

1. Reference #3 and #14 are incomplete and should be completed such as; a.#3. doi:10.5146/tjpath.2017.01408 b. #14. Histopathology 2002; 41(1):1-29.

2. The word "enterocoelia" or "enterocoelic mass" is not commonly used word in the medical literature, instead of it, "intraabdominal mass" is better.

3. In the Discussion section, "higher rate of replace" should be corrected by "higher rate of relapse".

4. Again in the Discussion section:

a."carcinogenic" in place of "carcinogenetic",

b."Castleman" in place of "castleman",

c. "Epstein-Barr" in place of "Epstein-barr",

d."interdigitating dendritic cell sarcoma (IDCS)" in place of "dendrite cell sarcoma (IDCS)" at the 2nd paragraph in pp 6.

e."recommended" in place of "commended" in pp 7.

After these corrections, this case report is worth publishing.

Thanks! We really appreciated that the review made such detailed comments and corrections to strengthen our manuscript. All the issues were taken cared and revised accordingly.